Author's response to reviews

Title: Progressive neurological disease induced by tacrolimus in a renal transplant recipient: Case report.

Authors:

Marjan Chegounchi (mchegoun@yahoo.com)
Michael G Hanna (m.hanna@ion.ucl.ac.uk)
Guy H Neild (g.neild@ucl.ac.uk)

Version: 3 Date: 26 February 2006

Author's response to reviews:

However, before acceptance, can you clarify Dr Nassogne's comment.

question: "are you sure that your picture is T1-MRI and T2?"

The question does not quite make sense. We are now only showing one figure, which is MRI T1 - as stated in the text and in the figure caption (legend).

There is one reference in the Discussion to T2 images, which is a quote from a cited reference.

I have not changed the submitted manuscript.